

Citation: Hannon E, Knox O, Sugden K, Burrage J, Wong CCY, Belsky DW, et al. (2018) Characterizing genetic and environmental influences on variable DNA methylation using monozygotic and dizygotic twins. PLoS Genet 14(8): e1007544. <u>https://doi.</u> org/10.1371/journal.pgen.1007544

Editor: John M. Greally, Albert Einstein College of Medicine, UNITED STATES

Received: February 5, 2018

Accepted: July 6, 2018

Published: August 9, 2018

Copyright: © 2018 Hannon et al. This is an open access article distributed under the terms of the <u>Creative Commons Attribution License</u>, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: Raw DNA methylation data are available in Gene Expression Omnibus (GEO) under accession number GSE105018. Estimates of the contribution of genetic and environmental influences to DNA methylation at all sites profiled in this study are available as a resource for the research community (http://www.epigenomicslab.com/online-dataresources/).

Funding: The E-Risk Study is funded by the Medical Research Council (G1002190). Additional

RESEARCH ARTICLE

Characterizing genetic and environmental influences on variable DNA methylation using monozygotic and dizygotic twins

Eilis Hannon¹, Olivia Knox¹, Karen Sugden², Joe Burrage¹, Chloe C. Y. Wong³, Daniel W. Belsky⁴, David L. Corcoran⁵, Louise Arseneault³, Terrie E. Moffitt^{2,3,5,6}, Avshalom Caspi^{2,3,5,6}, Jonathan Mill¹*

1 University of Exeter Medical School, University of Exeter, Exeter, United Kingdom, 2 Department of Psychology and Neuroscience, Duke University, Durham, NC, United States of America, 3 Social, Genetic & Developmental Psychiatry Centre, Institute of Psychiatry, Psychology & Neuroscience, King's College London, London United Kingdom, 4 Department of Population Health Sciences, Duke University School of Medicine, Durham, NC, United States of America, 5 Center for Genomic and Computational Biology, Duke University, Durham, NC, United States of America, 6 Department of Psychiatry and Behavioral Sciences, Duke University Medical School, Durham, NC, United States of America

* j.mill@exeter.ac.uk

Abstract

Variation in DNA methylation is being increasingly associated with health and disease outcomes. Although DNA methylation is hypothesized to be a mechanism by which both genetic and non-genetic factors can influence the regulation of gene expression, little is known about the extent to which DNA methylation at specific sites is influenced by heritable as well as environmental factors. We quantified DNA methylation in whole blood at age 18 in a birth cohort of 1,464 individuals comprising 426 monozygotic (MZ) and 306 same-sex dizygotic (DZ) twin pairs. Site-specific levels of DNA methylation were more strongly correlated across the genome between MZ than DZ twins. Structural equation models revealed that although the average contribution of additive genetic influences on DNA methylation across the genome was relatively low, it was notably elevated at the highly variable sites characterized by intermediate levels of DNAm that are most relevant for epigenetic epidemiology. Sites at which variable DNA methylation was most influenced by genetic factors were significantly enriched for DNA methylation guantitative trait loci (mQTL) effects, and overlapped with sites where inter-individual variation correlates across tissues. Finally, we show that DNA methylation at sites robustly associated with environmental exposures such as tobacco smoking and obesity is also influenced by additive genetic effects, highlighting the need to control for genetic background in analyses of exposure-associated DNA methylation differences. Estimates of the contribution of genetic and environmental influences to DNA methylation at all sites profiled in this study are available as a resource for the research community (http://www.epigenomicslab.com/online-data-resources).



support was provided by the National Institute of Child Health and Human Development (HD077482), a Distinguished Investigator Award from the American Asthma Foundation to JM, and by the Jacobs Foundation. LA is a Mental Health Leadership Fellow for the U.K. Economic and Social Research Council. Some of the work reported here used a high-performance computing facility partially supported by grant 2016-IDG-1013 from the North Carolina Biotechnology Center. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

Author summary

The study of monozygotic (MZ) and dizygotic (DZ) twins provides an opportunity for exploring the extent to which heritable and environmental factors contribute to phenotypic variation in human populations. We exploit the twin study design to explore the factors influencing epigenetic variation between individuals, focussing on DNA methylation, the best-characterized and most stable epigenetic modification. We find that site-specific levels of DNA methylation are more strongly correlated across the genome between MZ than DZ twins. While the average contribution of additive genetic influences on DNA methylation is relatively low, it is notably elevated at sites that are highly variable and have intermediate levels of DNAm, which are most relevant for epigenetic epidemiology. Sites at which variable DNA methylation is strongly influenced by genetic factors are enriched for DNA methylation quantitative trait loci (mQTL) effects, and overlap with sites where inter-individual variation correlates across tissues. Importantly, we show that DNA methylation at sites robustly associated with environmental exposures such as smoking and obesity is also influenced by genetic effects, highlighting the need to control for genetic background in analyses of exposure-associated DNA methylation differences. Finally, we present a searchable database cataloguing the genetic and environmental contributions to variable DNA methylation across the genome (http://www.epigenomicslab.com/onlinedata-resources).

Introduction

The study of twins provides an opportunity for exploring the extent to which heritable and environmental factors contribute to phenotypic variation in human populations [1]. By comparing concordance rates between monozygotic (MZ) and dizygotic (DZ) twins it has been shown that most human traits are, at least in part, influenced by DNA sequence variation [2]. The fact that genetically-identical MZ twins exhibit phenotypic differences indicates that non-sequence based factors, usually attributed to the environment, also contribute to phenotypic variation. Increasing knowledge about the biology of the genome has stimulated interest in the role of epigenetic processes—acting to developmentally regulate gene expression via modifications to DNA, histone proteins, and chromatin—in mediating phenotypic variation across the life-course. Growing evidence identifies epigenetic differences between MZ twins [3], and epigenetic variation is associated with a range of health and disease phenotypes [4].

The primary focus of epigenetic epidemiology is on DNA methylation, the best-characterized and most stable epigenetic modification, which is assumed to influence gene expression via the disruption of transcription factor binding and the attraction of methyl-binding proteins that initiate chromatin compaction and gene silencing. DNA methylation can be influenced by both environmental and genetic factors, meaning that careful study design in epigenomewide association studies (EWAS) is important to minimize the influence of confounders and false positives [4, 5]. There is evidence that certain exposures-for example, to tobacco smoke [6-8], dietary factors [9, 10] and psychosocial stress [11, 12]-are associated with changes in DNA methylation at specific sites across the genome. Likewise, studies have identified associations between DNA sequence variation and DNA methylation at sites across the genome [13-16]; these DNA methylation quantitative trait loci (mQTLs) often overlap with DNA variants associated with levels of gene expression (expression quantitative trait loci; eQTLs)[14, 17], providing a potential mechanism linking genetic variation to gene regulation. Researchers are starting to exploit the twin study design to further explore the extent to which epigenetic variation between individuals is influenced by genetic and environmental factors. Recent studies have shown that DNA methylation profiles are more similar between related individuals than unrelated individuals, with greater concordance between MZ than DZ twins [18, 19]. Twin studies suggest that the proportion of variance in DNA methylation explained by genetic factors is on average low (typically 5–19%) at the majority of sites that have been tested across the genome [19–21]. Importantly, however, the contribution of genetic and environmental factors to DNA methylation varies at sites across the genome, and potentially differs as a function of tissue, age and sex [21]. Studies investigating associations between DNA methylation and phenotypic variation, should not dismiss the impact that genetic variation may have on their results.

Here we report findings about the genetic and environmental architecture of DNA methylation in whole blood at age 18 years using samples collected from the Environmental Risk (E-Risk) Longitudinal Twin Study, a representative birth cohort of young-adult twins based in the UK. Young adulthood is a life stage when people show great variation in health risk behaviors and exposures that have been hypothesized to alter an individuals' epigenome. Our goal was to characterize the genetic and environmental determinants of variation in DNA methylation in order to inform future methylomic analyses of complex traits. By analyzing a sample where all twin pairs provided a whole blood sample at the same age, we minimize the confounding influence of age-associated variation.

We first used structural equation modeling to calculate the proportion of variance in DNA methylation explained by additive genetic (A), shared environmental (C) and unshared (or unique) environmental (E) factors at sites across the genome. Second, we explored whether the contribution of genetic and environmental influences on DNA methylation differs depending upon the level and/or variability in DNA methylation at individual sites. Third, we assessed how genetic and environmental influences on DNA methylation differ as a function of genic location, describing the factors influencing variable DNA methylation across gene regulatory regions. Fourth, we tested the hypothesis that sites characterized by highly heritable levels of DNA methylation are enriched for known mQTL effects. Fifth, we explored the extent to which biological phenotype estimates derived from DNA methylation data itself (e.g. age and blood cell proportions) are influenced by genetic and environmental factors, in addition to estimating the genetic and non-genetic contribution to levels of DNA methylation at sites robustly associated with specific environmental exposures (e.g. tobacco smoking and obesity). Finally, as a resource for the research community, we present a searchable database cataloguing the genetic and environmental contributions to variable DNA methylation across all sites on the Illumina 450K array (http://www.epigenomicslab.com/online-data-resources).

Results

Site-specific levels of DNA methylation are more strongly correlated between MZ twins than DZ twins, especially at sites with variable and intermediate levels of DNA methylation

We quantified genome-wide patterns of DNA methylation using the Illumina Infinium HumanMethylation450 BeadChip ("450K array") in DNA samples isolated from whole blood collected at age 18 years from members of the E-Risk cohort [22]. After implementing a stringent quality control (QC) pipeline (see <u>Methods</u>), our final sample included 426 MZ twin pairs (48.5% female) and 306 DZ twin pairs (49.2% female) (1,464 individuals, a representative 65.6% of participants, see <u>Methods</u>).



Fig 1. Monozygotic (MZ) twins are more concordant than dyzygotic (DZ) twins for DNA methylation at the majority of autosomal sites tested. (a) Scatterplot showing the correlation of DNA methylation values within MZ twin-pairs (x-axis) and DZ twin-pairs (y-axis) for all 420,857 autosomal Illumina 450K array sites passing our stringent quality control (QC) pipeline. MZ twin pairs are more similar to each other than DZ twin pairs for 277,077 (65.8%) of sites (sign-test P = 1.98x10-323). The elevated concordance of DNA methylation in MZ twins compared to DZ twins is more pronounced amongst both (b) "variable" sites (77.6%, sign-test P = 1.48x10-323) and (c) sites with intermediate levels of DNAm (83.0%, sign-test P = 9.88x10-324). The red diagonal line indicates x = y. The color indicates the density of data points ranging from yellow (highest) to grey (lowest).

We first assessed the profile of DNA methylation across all 420,857 autosomal 450K array sites included in our final dataset. As expected, these 'global' patterns of DNA methylation were highly stable between individuals (S1 Fig), although the average inter-individual correlation of DNA methylation across sites was significantly higher between siblings than between unrelated individuals (P = 2.20×10^{-223}). MZ twin pairs were more similar to each other than DZ twin pairs for the majority of sites tested (N = 277,077 (65.8%), sign test P = 1.98×10^{-323}) (Fig 1); the average sibling correlation across the 420,857 sites was significantly higher for MZ twin-pairs than for DZ pairs (mean MZ sibling correlation = 0.996, mean DZ sibling correlation = 0.995, P = 1.29×10^{-34}). The magnitude of this difference was relatively small, reflecting the fact that most autosomal 450K array probes are characterized by consistently high (>80%) or low (<20%) levels of DNA methylation, and minimal inter-individual variation. We therefore estimated sibling correlations for the subset of autosomal DNA methylation sites we defined as either "variable" (i.e. those where the range of DNA methylation values for the middle 80% of individuals was greater than 5%; N = 214,991 sites (51.1%)) or with intermediate levels of DNAm (i.e. those where the mean level of DNA methylation was between 20% and 80%; N = 131,728 sites (31.3%)) (see <u>Methods</u>). These probe subsets were not distinct; the majority (N = 127,935 (97.1%)) of DNA methylation sites with intermediate levels of DNAm were also classed as "variable" (S2 Fig). The elevated concordance in DNA methylation levels in MZ twins compared to DZ twins was more pronounced amongst both "variable" sites (number of sites at which MZ twin pairs are more similar to each other than DZ twin pairs = 166,783 (77.6%), sign test $P = 1.48 \times 10^{-323}$) and sites with intermediate levels of DNAm (number of sites at which MZ twin pairs are more similar to each other than DZ twin pairs = 109,303 (83.0%), sign test P = 9.88×10^{-324}) (Fig 1). Furthermore, there was an overall elevated average sibling similarity for DNA methylation levels in MZ twins compared to DZ twins amongst both "variable" DNA methylation sites (mean MZ sibling correlation = 0.989, mean DZ sibling correlation = 0.985, P = 3.92×10^{-38}) and DNA methylation sites with intermediate levels of DNAm (mean MZ sibling correlation = 0.979, mean DZ sibling correlation = 0.968, P = 1.55×10^{-39}) (<u>S1 Fig</u>), consistent with findings from previous twin studies of DNA methylation in whole blood [21, 23].

Autosomal DNA methylation is predominantly influenced by non-shared environmental factors

DNA methylation is widely hypothesized to be a mechanism by which both heritable and environmental factors can influence the regulation of gene expression and function, but little is known about the extent to which DNA methylation at specific sites is actually influenced by genetic and non-genetic factors. We fitted structural equation models to estimate the proportion of variance in DNA methylation explained by additive genetic effects (A), shared environmental effects (C) and unshared (or unique) environmental effects (E) across all 420,857 autosomal sites (see Methods) (Table 1). The average contribution of additive genetic effects across all DNA methylation sites was relatively low but highly variable (mean A = 15.9%(SD = 20.8%) (Fig 2A-2C); our mean estimate of heritability was slightly below that observed in previous studies of older and more variably-aged twin-pairs [19, 23]. On average, the largest contribution to variation in DNA methylation was attributable to unique environmental influences, which also indexes measurement error (mean E = 67.4% (SD = 22.9%)). The mean estimate for common environmental influences across all 420,857 autosomal sites was similar to that for additive genetic effects (mean C = 16.7% (SD = 17.8%)). These data highlight that variation in DNA methylation can be influenced by both genetic and non-genetic factors, and that the relative importance of these influences differs across sites in the genome. Because whole blood is a heterogeneous tissue, we derived blood cell proportion estimates for each sample using the DNAm data (see Methods) and repeated our structural equation modelling in an attempt to explore the effects of cellular heterogeneity on heritability estimates of DNAm. Including derived blood cell-types as a covariate in our model did not change the pattern of results (mean A = 16.5% (SD = 21.2%), mean C = 12.6% (SD = 13.7%), mean E = 71.0% (SD = 20.9%) (S3 Fig), with estimates for genetic and environmental influences on DNAm across sites being highly correlated across both models (S4 Fig). Fig 3 shows examples of sites at which the level of DNA methylation was influenced by a high (Fig 3A) and low (Fig 3B) additive genetic component. MZ and DZ twin correlations and estimates for A, C, and E for all Illumina 450K array sites are available as an online resource at http://www.epigenomicslab. com/online-data-resources).

Table 1. The contribution of additive genetic and environmental factors to levels of DNA methylation. Shown are the results from structural equation models to esti-
mate the mean proportion of variance in DNA methylation explained by additive genetic effects (A), shared environmental effects (C) and unshared (or unique) environ-
mental effects (E) across Illumina 450K probes. Results are presented separately for DNA methylation sites located on the autosomes and X-chromosome, and stratified by
whether they have intermediate levels of DNAm and/or are "variable".

		A		С		E		
Autosomes (all twins)								
	N probes	Mean	SD	Mean	SD	Mean	SD	
All	420,857	15.9%	20.8%	16.7%	17.8%	67.4%	22.9%	
Intermediate levels of DNAm	131,728	27.3%	24.6%	16.8%	16.5%	55.9%	22.3%	
Variable	214,991	23.0%	23.8%	15.9%	16.8%	61.1%	23.2%	
X Chromosome (female twins)								
All	9,896	30.2%	17.0%	13.6%	20.4%	56.3%	22.1%	
Intermediate levels of DNAm	7,911	32.1%	15.6%	14.8%	21.5%	53.1%	21.5%	
Variable	9,127	31.3%	16.5%	13.5%	20.5%	55.3%	21.7%	
X Chromosome (male twins)								
All	9,896	12.1%	17.5%	17.7%	24.5%	70.2%	25.6%	
Intermediate levels of DNAm	2,778	18.8%	20.7%	16.1%	16.7%	65.1%	20.0%	
Variable	5,377	15.0%	19.4%	15.5%	18.5%	69.4%	22.1%	

https://doi.org/10.1371/journal.pgen.1007544.t001



Fig 2. The proportion of variance in DNA methylation explained by additive genetic effects (A), shared environmental effects (C) and unshared (or unique) environmental effects (E) across autosomal sites. Panels a-c show density distributions for estimates of A, C, and E across all 420,857 autosomal DNA methylation sites. At the majority of autosomal sites, environmental factors contribute more to the observed variance in DNA methylation than additive genetic factors. We observe significantly higher average heritability estimates for DNA methylation across the subset of DNA methylation sites defined as "variable" (d-f) (mean A = 23.0% (SD = 23.8%); Mann Whitney P < $2.2x10^{-16}$) and (g-i) "sites with intermediate levels of DNAm" (mean A = 27.3% (SD = 24.6%); Mann Whitney P < $2.2x10^{-16}$).



Fig 3. Examples of autosomal sites at which DNA methylation is differentially influenced by additive genetic and environmental factors. (a) An example of a site (cg00002033) at which DNA methylation is highly heritable. The scatterplot shows DNA methylation values in MZ (left panel) and DZ (right panel) twin pairs. Each point represents an individual twin-pair. At this site, the correlation of DNA methylation is markedly higher in MZ twins (r = 0.882) compared to DZ twins (r = 0.484). Structural equation modelling highlights that DNA methylation at this site is strongly influenced by additive genetic effects (A = 79.7%, C = 8.53%, E = 11.8%). (b) An example of a site (cg00000289) at which DNA methylation is not strongly influenced by genetic factors. The scatterplot shows DNA methylation values in MZ (left panel) and DZ (right panel) twin pairs. Each point represents an individual twin-pair. At this site, the correlation of DNA methylation is similar in both MZ twins (r = 0.363) and DZ twins (r = 0.449) highlighting that DNA methylation is strongly influenced by the environment (A = 0%, C = 40.2%, E = 59.8%). MZ and DZ correlations for DNA methylation across all sites on the Illumina 450K array can be visualized at http://www.epigenomicslab.com/online-data-resources.

Additive genetic influences on DNA methylation are highest at highly variable sites and sites with intermediate levels of DNAm

We next tested the hypothesis that DNA methylation at sites which are "variable" or have intermediate levels of DNAm is more highly heritable than other sites in the genome. Average additive genetic influences on DNA methylation were markedly higher at "variable" autosomal sites compared to non-variable sites (mean A = 23.0% (SD = 23.8%), Mann Whitney $P < 2.2x10^{-16}$) (Fig 2D and S5 Fig). Likewise, additive genetic influences on DNA methylation were significantly higher at autosomal sites with intermediate levels of DNAm compared to



Fig 4. The contribution of genetic and environmental influences on DNA methylation at autosomal sites differs as a function of average DNA methylation level at that location. Shown are estimates of additive genetic effects (A), shared environmental effects (C) and non-shared (or unique) environmental effects (E) against mean DNA methylation level. The most heritable sites are characterized by intermediate levels of DNA methylation.

hyper/hypo-methylated sites (mean A = 27.3% (SD = 24.6%), Mann Whitney P < 2.2×10^{-16}), with a striking inverted U-shaped relationship between the level of DNA methylation and the extent to which it was influenced by additive genetic factors (Fig 2G and Fig 4). In contrast, the influence of non-shared environmental factors was significantly lower at "variable" autosomal sites compared to non-variable sites (mean E = 61.1% (SD = 23.2%); Mann-Whitney $P < 2.2 x 10^{-16}$) (Fig 2F). The contribution of non-shared environmental factors was also lower at autosomal sites with intermediate levels of DNAm compared to either hyper- or hypo-methylated sites (mean E = 55.9% (SD = 22.3%); Mann-Whitney P $< 2.2 \times 10^{-16}$) (Fig 2I); there is a U-shaped relationship between the mean level of DNA methylation and the proportion of variance explained by unique environmental effects; the smallest contribution of E was observed at sites that were 56-58% methylated (Fig 4). Shared environmental influences were fairly stable and not strongly affected by either the average variability or level of DNA methylation. These results are important because they suggest that the effects of genetic variants associated with phenotypic differences are likely to be more pronounced at DNA methylation sites that are variable or have intermediate levels of DNAm compared to hypo- or hyper methylated sites, which are more stable in the population and often associated with cell-type-specific patterns of gene expression.

Genetic influences on DNA methylation are not evenly distributed across genic regions

Although DNA methylation across CpG-rich promoter regions is often associated with the repression of gene expression, recent work has revealed a more nuanced relationship between DNA methylation and transcription that is frequently dependent on genomic context [24].

DNA methylation in the gene body, for example, can be a marker of active gene transcription [25, 26], potentially playing a role in regulating alternative splicing and isoform diversity. Given these contextual differences, we tested whether genetic and environmental contributions to variable DNA methylation differ across genomic domains. As DNAm sites located in specific gene features differ in their variability, these analyses focused on our subset of "variable" DNAm sites to prevent any potential confounding. First, we used a sliding-window approach to examine how the proportion of variation in DNA methylation explained by genetic and environmental influences changes across a canonical gene region (S6 Fig). There was a peak in the contribution of shared environmental influences in the vicinity of the transcription start site (TSS), accompanied by a reduction in the contribution of non-shared environmental influences. The contribution of additive genetic factors to variation in DNA methylation was highest at sites located immediately upstream of the TSS, and also in a region spanning ~5 kilobases (kb) downstream of the transcription termination site. Second, we tested the extent to which DNA methylation levels at sites annotated to specific genic features (S7 Fig) and CpG island features (S8 Fig) were influenced by additive genetic or environmental factors. Variation in DNA methylation at sites in the immediate vicinity of a TSS, or annotated to a first exon or CpG island, were associated with significantly higher additive genetic and shared environmental influences (all Mann-Whitney $P < 2.2 \times 10^{-16}$) (<u>S1 Table</u>). Given the presumed importance of promoter-region DNA methylation in regulating gene expression, these observations suggest that both environmental and genetic factors can influence transcriptional regulation via DNA methylation at these promoter-region locations and that, on average, the effects across features are relatively consistent. Finally, we investigated how the influence of genetic and environmental factors on DNAm varies across regulatory features and chromatin states defined by ChromHMM using ENCODE ChIP-seq data for a well-characterized lymphoblastoid cell line (GM12878) (S9 Fig). This analysis revealed higher levels of additive genetic effects on DNAm at sites in insulators (mean A = 23.0%, SD = 24.2%), repressed (mean A = 19.6%, SD = 21.4%) and repetitive/CNV regions (mean A = 24.8-27.0%, SD = 25.8-26.2%, with moderate levels of heritability in enhancer regions (mean A = 17.5-19.1%, SD = 20.9 - 22.1%). In contrast, DNAm at sites located in promoters is characterized by an increased proportion of variance explained by unique environmental factors (E = 65.6-67.8%, SD = 22.3–23.2%) reflecting the genic annotation results above.

Sites at which variable DNA methylation is strongly influenced by additive genetic factors are significantly enriched for mQTL effects

Given that epigenetic epidemiology aims to understand both the causes and phenotypic consequences of differential DNA methylation, we focused our subsequent analyses on the subset of 214,991 "variable" autosomal DNA methylation sites. Hypothesizing that the majority of heritable DNA methylation sites identified in this study are influenced by common genetic variation, we tested whether they were enriched for mQTL effects, i.e. common genetic variants known to be robustly associated with DNA methylation at specific sites [13, 27, 28]. We used a large mQTL database generated by our group on an independent set of whole blood samples [29] to identify overlap with the most highly heritable DNA methylation sites (defined as those with A > 0.8; n = 4,882) identified in the E-Risk cohort. DNA methylation at 84.7% of these sites was significantly associated with at least one common genetic variant using a stringent mQTL threshold ($P < 1x10^{-8}$) (S2 Table); this represented a highly significant enrichment for mQTL effects ($P < 2.2x10^{-16}$) compared to less-heritable DNA methylation sites (defined as those with A < 0.8), amongst which only 24.5% were associated with a mQTL variant. Of note, mQTL effect sizes vary as a function of the mean level of DNAm. Sites with intermediate levels of DNAm are associated with larger mQTL effects (mean = 4.99% change in methylation per allele (SD = 3.61%)) compared to sites characterized as being hyper- or hypo-methylated (mean = 3.56% change in methylation per allele (SD = 2.79%); Mann-Whitney P < $2.2x10^{-16}$); this parallels the relationship observed between the level of DNAm and the influence of additive genetic factors (S10 Fig). These findings suggest that the incorporation of common SNP data into epigenome-wide association studies (EWAS) will facilitate understanding about the contribution of genetic and non-genetic factors to trait-associated methylomic variation. An example of a highly heritable DNA methylation site (cg02573566, A = 96.9%) that was also associated with an mQTL SNP (rs11548104, $P = 5.95 \times 10^{-179}$) is shown in <u>S11 Fig</u>. Of note, observed DNA methylation at highly heritable sites for which we did not detect an mQTL (15.3%) does not necessarily signal false positives as these sites may be associated with rare variation or larger structural variants not assessed in existing mQTL databases. mQTLs influencing levels of DNA methylation at highly heritable sites were associated with larger effects (mean change in DNA methylation per allele = 6.77% (SD = 4.48%)) compared to all identified mQTLs (mean change in DNA methylation per allele = 3.03% (SD = 3.10%)) (P = $< 2.2x10^{-16}$). Across all autosomal 450K array sites, there was a relatively linear relationship between the contribution of genetic influences to variation in DNA methylation and the proportion of sites influenced by an mQTL (\$12 Fig). In contrast, the proportion of DNA methylation sites that were associated with an mQTL decreased as the contribution of the common or unique environment to levels of DNA methylation increased. Taken together, these findings confirm our hypothesis that DNA methylation at the majority of highly heritable sites is directly influenced by common genetic variants.

DNA methylation sites at which inter-individual variation is correlated across tissues are characterized by higher levels of heritability

Epigenetic association studies of phenotypes where the presumed tissue of interest is challenging to obtain (e.g. regions of the human brain) typically use more accessible peripheral tissues (e.g. whole blood) under the premise that variation identified in these 'proxy' tissues potentially mirrors that in the disease-relevant tissue. We have previously shown, however, that whole blood generally has limited utility for inferring inter-individual variation in multiple regions of the human brain [30]. Where there is significant co-variation between two tissues from the same individual, we hypothesized that this is likely to reflect genetic effects on DNA methylation that are manifest across tissues. We used the matched blood and brain DNA methylation datasets, previously generated by our group [30], to confirm that DNA methylation at sites characterized by high inter-individual co-variation across tissues from the same individual is more likely to be influenced by heritable factors. For example, we observed a striking increase in the heritability of DNA methylation at the subset of sites at which interindividual variation in our prior sample was strongly correlated between whole blood and the prefrontal cortex (covariation between blood and prefrontal cortex > 0.5, N = 9,212 sites) compared to those at which variation was less correlated across tissues (median A = 71.1% vs 14.7%, Mann-Whitney P $< 2.2 \times 10^{-16}$) (Fig 5A). Overall, there was a strong positive correlation (r = 0.500) between the additive genetic contribution to DNA methylation and tissue co-variation (blood vs prefrontal cortex) across variably methylated sites (S13 Fig), confirming that sites at which DNA methylation co-varies across tissues are more likely to be influenced by heritable factors. Similar effects were seen for the other brain regions profiled from the same individual donors (entorhinal cortex, superior temporal gyrus and cerebellum). An example of a site where DNA methylation significantly covaries between whole blood and brain, and is



Fig 5. DNA methylation sites at which inter-individual variation is correlated across tissues are characterized by higher levels of heritability. (a) A density plot of heritability estimates for DNA methylation at sites split by the extent to which DNA methylation co-varies between whole blood and the prefrontal cortex using data from Hannon et al (2015). Heritability is significantly higher in probes where the cross-tissue covariation in DNA methylation is high ($r^2 > 0.5$, red). (**b**-**h**) An example of a probe (cg08449049) at which DNA methylation is strongly influenced by additive genetic effects and also co-varies between blood and multiple regions of the human brain. Shown are scatterplots of DNA methylation values at cg08449049 for (**b**) MZ (corr = 0.851) and **c**) DZ (corr = 0.364) twin pairs. Each point represents an individual twin-pair. (**d**) A boxplot of the distribution of DNA methylation levels at cg08449049 in blood and four brain regions (PFC = prefrontal cortex, EC = entorhinal cortex, STG = superior temporal gyrus, CER = cerebellum) from the same individual donors using data generated by Hannon et al (2015). (**e**-**h**) Scatterplots of the DNA methylation values in blood against the DNA methylation values in each of the four brain regions showing that there is significant covariation across tissues.

strongly influenced by additive genetic effects, is shown in Fig 5B–5H. These results are important because they suggest that concerns regarding tissue-specific effects on DNA methylation are likely to be more relevant for studies of environmentally-induced variation as compared to analyses of genetic influences on DNA methylation.

Genetic influences on DNA methylation at sites on the X chromosome are also highest at sites characterized by intermediate levels of DNAm and high variability

Because DNA methylation on the X-chromosome differs markedly between males and females-primarily due to its role in regulating the dosage compensation of X-linked genes (see <u>\$14 Fig</u>)—the analyses presented above focused solely on autosomal DNA methylation sites. We next estimated the proportion of variance in DNA methylation explained by additive genetic effects, shared environmental effects and non-shared (or unique) environmental effects for probes on the X chromosome in male and female twins separately (male: 156 DZ twin pairs, 219 MZ twin pairs; female: 150 DZ twin pairs, 207 MZ twin pairs) (Table 1). As hypothesized, X-chromosome DNA methylation was much more variable in females than males; the majority (N = 9,127, 92.2%) of X-linked DNA methylation sites met our criteria for being "variable" in females compared to just over half (N = 5,377, 54.3%) in males. Most DNA methylation sites classified as "variable" in males were also found to be "variable" in females (N = 5, 195; 96.6%). In males, the contribution of genetic and environmental influences to DNA methylation at sites on the X-chromosome was similar to that observed at autosomal loci; for males, more variation was attributed to unique environmental influences (mean = 69.4%, SD = 22.1%) than shared environmental (mean = 15.5%, SD = 18.5%) or additive genetic (mean = 15.0%, SD = 19.4%) influences (<u>515 Fig</u>). Furthermore, the influence of additive genetic factors on male X-chromosome DNA methylation was highest at sites characterized by either "intermediate levels of DNAm" (S16 Fig) or "variable" levels of DNA methylation (\$17 Fig). Although most variance in X-chromosome DNA methylation in females could also be attributed to the unique environment (mean E =55.3%, SD = 21.7%), the average contribution of additive genetic factors (mean A = 31.3%, SD = 16.5%) was significantly higher compared to that observed at autosomal sites (P < 2.2×10^{-16}) and X-linked sites in males (P < 2.2×10^{-16}) (Table 1). While the influence of genetic and environmental factors on DNA methylation across sites on the X-chromosome was positively correlated between males and females (S18 Fig), with the strongest correlation seen for unique environmental influences (r = 0.381), there was some notable heterogeneity. A number of sites, for example, were characterized by sex-specific additive genetic influences on DNA methylation (S19 Fig and S20 Fig). These results are interesting as they could potentially mediate observed sex differences for certain inherited phenotypes. This heterogeneity of effects may also have negative effects on power for statistical significance in EWAS analyses that combine males and female samples to analyze sites on the X chromosome; to truly disentangle genetic and environmental effects on X-chromosome DNA methylation it is important to analyze the sexes separately. Finally, we examined the genetic and environmental contribution to variable DNAm across regions annotated to the small subset of genes known to escape X-chromosome inactivation (XCI) in females. Using RNA-seq data from the GTEx consortium [31] we selected DNAm sites annotated to the 5'UTR or within 1500 bp of the transcription start site of genes highlighted as escaping XCI. As expected, the distribution of DNAm across sites annotated to genes escaping XCI is dramatically different to other X-chromosome sites in females, with a striking enrichment of hypomethylated loci. Despite the differences in levels of DNAm associated with genes escaping XCI, the contribution of additive genetic and environmental influences on DNAm at these sites is broadly comparable to that seen at sites across the X-chromosome in females (S21 Fig).

Estimates of chronological age and blood cell proportions derived from DNA methylation data are influenced by both genetic and environmental effects

A number of classifiers can be used to derive estimates of biological phenotypes including age (DNAmAge) [32] and the proportion (or abundance) of different cell types present in whole blood [32-34] from DNA methylation data. These estimates are useful because they can be incorporated as covariates in EWAS analyses when empirical measures are missing, or used as interesting variables in their own right in epidemiological analyses [35-37]. We examined the twin correlations for each of these derived variables (S22 Fig) and estimated the contribution of additive genetic and environmental influences to these measures by comparing MZ and DZ twins (<u>\$23 Fig</u>). The mean predicted DNAmAge of samples from participants in this study was 20.7 years (SD = 4.10 years), slightly higher and more variable that the actual age at sampling (mean = 18.4 years; SD = 0.37 years). As DNAmAge is associated with actual chronological age, age acceleration is typically calculated as the residual from a linear regression model of predicted age against reported age. Although the limited age variation in our sample provides limited power for structural equation modelling, we found that DNAmAge acceleration was characterized by an additive genetic contribution of 36.7%, with 42.8% and 20.5% of the variance explained by common environmental and unique environmental influences, respectively. This heritability estimate is lower than the 100% reported previously for age acceleration in a smaller set of newborns but comparable to the 39% reported for adult twin pairs (45-75 years old) [32]. The contribution of additive genetic and environmental influences differed dramatically across the predicted cellular heterogeneity variables, with heritability estimates ranging from 0% (for CD8 T cells and granulocytes) to 47.0% (for CD8+CD28-CD45RA- T cells) (S3 Table). For seven of the ten derived cell estimates, the largest proportion of variance was attributed to the influence of unique environmental factors. B cells had the largest proportion of variance estimated as being explained by common environmental factors (52.1%), and naïve CD8 T cells and natural killer cells had the largest proportion explained by genetic factors (at 42.1% and 40.0%, respectively). Comparison between these results and those for empirically-measured cell abundance data is not straightforward as in many cases the estimated cellular composition represents a proportion rather than abundance. Although, there is contradictory evidence in the literature about whether variation in specific blood cell types is more influenced by genetic or environmental factors [38-41], our results are consistent with reports that T cells have higher heritability estimates than B cells [38, 41].

DNA methylation at sites robustly associated with exposure to tobacco smoking and body mass index (BMI) is strongly influenced by additive genetic effects

Several environmental exposures have been robustly associated with differences in DNA methylation at specific sites across the genome, although the extent to which these relationships are potentially confounded by genetic influences is not known. We first examined whether variation in DNA methylation at sites associated with tobacco smoking—an exposure known to be characterized by robust and reproducible effects on DNA methylation [6, 7, 42, 43]–is also influenced by additive genetic factors. Using the extended E-Risk dataset including singletons (i.e. individuals whose co-twin did not contribute to our DNA methylation dataset), we performed an EWAS of tobacco smoking, identifying 97 differentially methylated positions (DMPs) (P < $1x10^{-7}$) (S4 Table) that are highly consistent with findings from previous studies of smoking in adults [44] (S24 Fig). We next examined the extent to which DNA methylation at these sites was influenced by genetic and environmental factors. We identified a strong genetic component to levels of DNA methylation at smoking-associated DMPs; overall there were significantly higher contributions of additive genetic influences (mean A = 37.7% (SD = 22.2%); Mann-Whitney $P = 3.20 \times 10^{-12}$) as well as shared environmental influences (mean C = 23.5% (SD = 16.0%); Mann-Whitney P = 0.00419) across smoking-associated DMPs compared to all "variable" DNA methylation sites, with a significantly smaller contribution of unique environmental influences (mean E = 38.9% (SD = 17.4%); Mann-Whitney P = 5.47×10^{-16}) (Fig 6). We next attempted to control for the fact that smoking behavior (and therefore the "exposure" itself) is a heritable trait [45, 46]; by only considering 18-year-old twin pairs where both members have never smoked it can be assumed that the influence of tobacco exposure on DNA methylation is negligible and any observed heritability at these sites cannot result from smoking. For 95 of 97 smoking-associated DMPs, the correlation of DNA methylation in MZ concordant nonsmokers (N = 315 twin-pairs) was greater than in DZ concordant non-smokers (N = 187 twin pairs) (Fig 6), representing a significant enrichment (P = 6.00×10^{-26}). S25 Fig highlights two DMPs at which DNA methylation was strongly associated with smoking status (cg05575921: $P = 1.73 \times 10^{-80}$; cg26703534: $P = 1.39 \times 10^{-90}$) but also was notably more correlated in MZ twin pairs (cg05575921: r = 0.845; cg26703534 r = 0.658) than DZ twin pairs (cg05575921: r = 0.579;



Fig 6. DNA methylation at sites associated with tobacco smoking is strongly influenced by additive genetic factors. Shown is a series of density plots for estimates of (a) additive genetic effects (A), (b) shared environmental effects (C) and (c) non-shared environmental effects (E) at 97 differentially methylated positions (DMPs) associated with smoking (green). Also shown are density plots for A, C and E at 'background' sites not associated with smoking (red). Shown below is a series of scatterplots showing the correlation in DNA methylation between MZ twins (x-axis) against DZ twins (y-axis) for sites associated with smoking in (d) all twins, (e) concordant non-smokers (n = 503 twin-pairs), (f) twins discordant for smoking status (n = 123 twin-pairs) and (g) concordant smokers (n = 106 twin-pairs). The shaded area on each plot indicates the heritability estimate (using Falconer's formula) for each site.

https://doi.org/10.1371/journal.pgen.1007544.g006

cg26703534: r = 0.444). These data are important because they provide evidence that smoking effects are not necessarily independent of smokers' genetic background, and that it is important to control for genetic background when testing for effects of tobacco on health. We also explored the genetic and environmental contributions to variation in DNA methylation at DMPs robustly associated with BMI[47], again observing that these had significantly higher additive genetic influences (mean A = 31.4% (SD = 19.4%); Mann-Whitney P = 1.83×10^{-11}) and shared environmental influences (mean C = 23.4% (SD = 15.4%)); Mann-Whitney P = 2.16×10^{-13}) compared to all "variable" DNA methylation sites (S26 Fig; S5 Table). These data highlight how DNA methylation at sites robustly associated with extrinsic factors can also be under strong genetic control, highlighting the need to control for genetic background in future EWAS analyses of exposure-associated DNA methylation differences.

Discussion

We quantified genome-wide patterns of DNA methylation in whole blood in 18-year-old young adults using samples collected from a large representative birth cohort of MZ and same-sex DZ twin pairs. We show that site-specific levels of DNA methylation are more strongly correlated between MZ twins than DZ twins, especially at sites with variable and intermediate levels of DNA methylation. Using structural equation models, we calculated the proportion of variance in DNA methylation explained by additive genetic effects, shared environmental effects and unshared (or unique) environmental effects, finding that, on average, the largest contribution to variation in DNA methylation can be attributed to unique environmental influences. Although the average contribution of additive genetic influences on DNA methylation was found to be relatively lower, it is variable and notably elevated at DNAm sites that are highly variable and have intermediate levels of DNAm. Interestingly, sites at which variable DNA methylation is strongly influenced by additive genetic factors are significantly enriched for blood mQTL effects, and also for sites at which inter-individual variation is correlated across tissues. Finally, we show that DNA methylation at sites robustly associated with exposures such as tobacco smoking and BMI is, in fact, also influenced by additive genetic effects, implying that environmental epigenetics research should routinely control for genetic background in future analyses. Estimates of the contribution of genetic and environmental influences to DNA methylation at all sites profiled in this study are available as a resource for the research community (http://www.epigenomicslab.com/online-data-resources).

Unlike previous studies that have used twins to explore the genetic and environmental architecture of DNA methylation [19, 21, 23], we focused solely on same-sex twins who were all the same chronological age, enabling us to negate the effects of age and DZ twin sex-discordance on variable DNA methylation. Despite these strengths, however, our study has a number of important limitations that should be considered. First, because our analyses focused solely on a cross-section of young adults we cannot say anything about how genetic and environmental influences on DNA methylation change over time. Of note, our average estimate of additive genetic influences on DNA methylation is slightly below that observed in previous studies of older and more variably-aged twin-pairs [19, 23]. Second, our study cohort comprised individuals of European descent, like most other studies into the causes of variable DNA methylation. We know, however, that there are important racial and socioeconomic inequalities in pathogenic exposures and it is crucial that future work explores the contribution of genetic and environmental contributions to epigenetic variation in non-Caucasian populations. Third, although the Illumina 450K array quantifies DNA methylation at sites annotated to the majority of genes, the actual proportion of sites across the genome interrogated by this technology is relatively low, with a predominant focus on CpG-rich promoter regions. It will

be important for future studies to explore factors influencing levels of DNA methylation across regions not well-covered by the Illumina 450K array, especially given our finding that genetic and environmental influences on DNA methylation are not evenly distributed across genic regions. Of note, most of the content (> 90%) of the Illumina 450K array is present on the new Illumina EPIC array [48] and the results presented here are therefore applicable to future studies using this technology. Fourth, our study only assessed a single tissue–whole blood–which itself is comprised of a heterogeneous mix of different cell-types. Although blood cell-type proportions can be accurately derived from whole blood DNA methylation data, it is likely that the contribution of genetic and environmental factors to methylomic variation differs across different cell-types. Future work should extend these analyses to quantify DNA methylation in purified blood cell-types and cell isolated from other tissues from MZ and DZ twins to explore the extent to which our findings are generalizable across tissues and cell-types. Of note, DNA methylation sites at which inter-individual variation is correlated across tissues were characterized by higher heritability, suggesting that genetic effects on DNA methylation may be relatively conserved across tissues and cell types.

Although the largest contributor to inter-individual variation in DNA methylation across all tested sites was found to be non-shared environmental factors, which also captures measurement error, our findings highlight the importance of genetic influences on DNA methylation. Genetic influences appear to be especially important in mediating levels of DNA methylation at highly variable DNA methylation sites and those that are characterized by high levels of covariation across tissues suggesting that concerns relating to tissue-specific effects may be less relevant for genetic studies of DNA methylation. As expected, sites at which variable DNA methylation is strongly influenced by additive genetic factors are significantly enriched for known mQTL effects. Our results could be potentially used to improve the power of mQTL studies by providing a refined list of 'heritable' DNA methylation sites, thereby reducing the multiple testing burden and sample sizes needed to identify significant mQTL associations. The mean estimate of shared environmental effects on DNAm across the genome was higher than previously reported [21] and comparable to the magnitude of influence of additive genetic factors. Given the young and comparable ages of the participants in the E-Risk cohort (all ~ 18 years old) it is plausible that a higher proportion of environmental influences are shared between the twins compared to the variably-aged and older twin pairs profiled in other studies.

To conclude, we have characterized the genetic and environmental architecture of methylomic variation in a large sample of young adult MZ and DZ twins. We show that both heritable and non-genetic factors influence DNA methylation in a site-specific manner, with the contribution of genetic variation being highest at the most variable DNA methylation sites. Social-science and health researchers in search of evidence for environmental effects on the genome should not assume that "epigenetic" equates to "environmental". Importantly, DNA methylation at sites robustly associated with extrinsic factors such as smoking and BMI can also be under strong genetic control. Our online database provides estimates of the extent to which variable DNA methylation across all sites profiled in this study are under genetic influence. Although this resource is limited by some of the features of this study–i.e. it focuses on individuals of European descent, a single age-group, and sites on the Illumina 450K array–it provides a useful framework for interpreting the results of epigenetic epidemiological studies undertaken in whole blood.

Materials and methods

Ethics statement

The study was approved by the NRES Committee London—Camberwell St Giles Ethics Committee, and The Joint South London and Maudsley and the Institute of Psychiatry Research Ethics Committee approved each phase of the E-Risk study (reference number: 1997/122). Parents gave written informed consent and twins gave oral assent between 5–12 years and then written informed consent at age 18.

Samples

Participants were members of the Environmental Risk (E-Risk) Longitudinal Twin Study, which tracks the development of a 1994–95 birth cohort of 2,232 British children[22]. Briefly, the E-Risk sample was constructed in 1999–2000, when 1,116 families (93% of those eligible) with same-sex 5-year-old twins participated in home-visit assessments. This sample comprised 56% monozygotic (MZ) and 44% dizygotic (DZ) twin pairs; sex was evenly distributed within zygosity (49% male). The study sample represents the full range of socioeconomic conditions in Great Britain, as reflected in the families' distribution on a neighborhood-level socioeconomic index (ACORN [A Classification of Residential Neighbourhoods], developed by CACI Inc. for commercial use)[49]: 25.6% of E-Risk families live in "wealthy achiever" neighborhoods; 29.6% vs. 26.9% in "comfortably off" neighborhoods; 13.4% vs. 13.9% in "moderate means" neighborhoods; and 26.1% vs. 20.7% in "hard-pressed" neighborhoods. E-Risk underrepresents "urban prosperity" neighborhoods because such households are often childless.

Home visits were conducted when participants were aged 5, 7, 10, 12 and most recently, 18 years (93% participation). Our epigenetic study used DNA from a single tissue: whole blood. At age 18, whole blood was collected in 10mL K2EDTA tubes from 1,700 participants and DNA extracted from the buffy coat. (Study members who did not provide blood provided buccal swabs, but these were not included in our methylation analysis to avoid tissue-source confounds). There were no differences between participants who did versus did not participate and who did versus did not provide blood in terms of their socioeconomic background, IQ, mental health, or victimization experiences [50].

Genome-wide quantification of DNA methylation

We assayed 1669 blood samples (out of 1700); 31 samples were not useable (e.g., due to low DNA concentration). ~500ng of DNA from each sample (diluted to a standard concentration of 25ng/µL) was treated with sodium bisulfite using the EZ-96 DNA Methylation kit (Zymo Research, CA, USA). DNA methylation was quantified using the Illumina Infinium Human-Methylation450 BeadChip ("Illumina 450K array") run on an Illumina iScan System (Illumina, CA, USA). Twin pairs were randomly assigned to bisulfite-conversion plates and Illumina 450K arrays, with siblings processed in adjacent positions to minimize batch effects. Data were imported using the methylumIDAT function in methylumi[51] and subjected to quality control analyses, checking for sex mismatches, genotype data that did not concur with those typed on Illumina OmniExpress24v1.2 arrays, and excluding low intensity samples (details in [50]). In total, samples from 1658 participants passed our QC pipeline. Data were processed with the *pfilter* function from the *wateRmelon* package[52] excluding 0 samples with >1% of sites with a detection p value >0.05, 567 sites with beadcount <3 in 5% of samples and 1448 probes with >1% of samples with detection p value >0.05. The data were normalized with the *dasen* function from the wateRmelon package [52]. This article reports about 732 complete twin pairs (426 MZ and 306 same-sex DZ). Prior to any analyses, probes with common (>5% MAF) SNPs within 10 bp of the single base extension and probes with sequences previously identified as potentially hybridizing to multiple genomic loci were excluded [53, 54], resulting in a final dataset of 430,802 probes. Zygosity of twin pairs in the E-Risk cohort was confirmed in two ways. First, signal intensities at the 65 SNP probes on the Illumina 450K array were used to confirm

that MZ twins were genetically identical. Second, SNP array genotype data for these samples was used to confirm that MZ twins shared 100% of their genetic variation (PI_HAT = 1) and DZ twins shared ~ 50% of their genetic variation (PI_HAT ~ 0.5). The results from these two stages were then cross-validated for final confirmation.

Structural equation modelling to estimate the contribution heritable and environmental influences on DNA methylation

Biometrical modelling was applied to every probe passing QC on the Illumina 450K array. Specifically, an ACE model was fitted to calculate the proportion of variance in DNA methylation explained by additive genetic (A), shared environmental (C) and unshared or unique environmental (E) factors, the latter which also includes measurement error. The assumptions behind this model are that additive genetic factors are perfectly correlated between MZ twins (i.e. genetic correlation = 1) but are only 50% correlated between DZ twins (i.e. genetic correlation = 0.5) and that shared non-heritable influences are equally similar between MZ and DZ twin pairs. The model was fitted using structural equation modelling implemented with functions from the OpenMx R package [55, 56]. For DNA methylation sites located on the autosomes this model was fitted using all twin pairs; for sites located on the X chromosome, the analysis was performed separately for males and females. Given the sparse coverage on the Y chromosome, Y-linked sites were dropped from analysis. The same model was used to calculate A,C and E estimates for the predicted age and cell composition variables generated with the Epigenetic Clock software[32].

Probe annotation of sites in the 450K array

The location of DNA methylation sites within genic features (5'UTR, 3'UTR, 1st Exon, gene body, within 200 or 1500bp of the transcription start site [TSS] and CpG island categories [CpG Island, shelf, shore]) were taken from the annotation files provided by Illumina (<u>ftp://ussd-ftp.</u> illumina.com/downloads/ProductFiles/HumanMethylation450/HumanMethylation450_15017482_v1-2.csv).

DNA methylation quantitative trait loci

DNA methylation quantitative trait loci (mQTL) were taken from a previously published study based on whole blood profiles from 639 adult samples [29]. After testing all DNA methylation sites against all genetic variants, 8,960,441 mQTL were identified using a p value threshold of 1×10^{-10} . From this set of mQTL, 98,239/389,246 (25.2%) of DNA methylation sites overlapping with the heritability analysis had an mQTL.

DNA methylation sites associated with tobacco smoking

To identify DNA methylation sites associated with tobacco smoking, a linear regression model was fitted across the extended E-Risk sample including singletons (n = 1,658). Current smokers (N = 392) were compared against former (N = 42) and never smokers (N = 1,223) whilst controlling for sex, batch, and 7 estimated variables relating to cellular heterogeneity generated with either the Houseman algorithm [33, 34] or Horvath Epigenetic clock [32]. To control for the fact that many members of the sample are related robust standard errors were calculated with the R packages *plm* [57] and sandwich [58] and used to generate p-values. 97 DNA methylation sites were associated with current smoking status at an experiment-wide p-value threshold of $1x10^{-7}$. It should be noted that the exact number of genome-wide significant associations for tobacco smoking differs slightly from that reported in [50] due to differences in

methods used to account for related samples and due to filtering DNA methylation sites based on their variability.

DNA methylation sites associated with BMI

DNA methylation sites associated with BMI were identified from the supplementary material published as part of the EWAS performed by Wahl et al [47]. Taking their 187 replicated, sentinel associations, 176 of these were present in our set of variable DNA methylation sites and therefore were included for comparison with our estimates of heritability.

Supporting information

S1 Table. The extent to which DNA methylation levels at sites annotated to specific genic features and CpG island features are enriched for the influence of additive genetic or environmental factors.

(PDF)

S2 Table. Sites at which DNA methylation is strongly influenced by additive genetic effects are often associated with mQTL variation. (XLSX)

S3 Table. Estimated contribution of additive genetic and environmental influences to estimated age and blood cell proportion estimates derived from DNA methylation data. (PDF)

S4 Table. Estimates of additive genetic and environmental influences on levels of DNA methylation at the 97 differentially methylated positions ($P < 1x10^{-7}$) associated with tobacco smoking. (PDF)

S5 Table. Estimates of additive genetic and environmental effects on levels of DNA methylation at 176 differentially methylated positions associated with BMI. (PDF)

S1 Fig. Genome-wide patterns of DNA methylation are highly correlated between siblings, with significantly higher average similarity in monozygotic (MZ) twin-pairs than dizygotic (DZ) twin-pairs. Shown are violin plots for the average correlations of DNA methylation within each sibling pair (stratified by relatedness) averaged across A) all autosomal DNA methylation sites (n = 420,857), B) autosomal sites characterized by "variable" DNA methylation (n = 214,991), C) autosomal sites characterized by "non-variable" DNA methylation (n = 205,866), D) autosomal sites with intermediate levels of DNA methylation (n = 131,728), and E) autosomal sites characterized as being either hypo- or hyper-methylated (n = 289,129). P-values are from a t-test comparing average correlations observed in MZ twins to those observed in DZ twins. Also shown are comparisons between random pairs of unrelated individuals selected from the E-Risk cohort.

(PDF)

S2 Fig. There is considerable overlap between the set of autosomal DNA methylation sites defined as being 'variable' and having intermediate levels of DNA methylation. (TIFF)

S3 Fig. The proportion of variance in DNA methylation explained by additive genetic effects (A), shared environmental effects (C) and unshared (or unique) environmental effects (E) across autosomal sites after adjusting for cellular composition. Panels a-c show

density distributions for estimates of A, C, and E across all 420,857 autosomal DNA methylation sites. At the majority of autosomal sites, environmental factors contribute more to the observed variance in DNA methylation than additive genetic factors. We observe significantly higher average heritability estimates for DNA methylation across the subset of DNA methylation sites defined as "variable" (**d**-**f**) (mean A = 29.3% (SD = 25.0%); Mann Whitney $P < 2.2x10^{-16}$) and (**g**-**i**) sites with intermediate levels of DNA methylation (mean A = 24.3% (SD = 24.2%); Mann Whitney $P < 2.2x10^{-16}$). (PDF)

S4 Fig. The contribution of genetic and environmental influences on DNA methylation is not strongly influenced by blood cell heterogeneity. Scatterplots of additive genetic effects (A), shared environmental effects (C) and non-shared (or unique) environmental effects (E) for all autosomal DNA methylation sites (n = 420,857), comparing DNA methylation data unadjusted for cellular composition (x-axis) and DNA methylation data adjusted for cellular composition variables (y-axis). Each point represents a DNA methylation site and the colour of the point indicates the density of points at that location (gray–low to yellow–high). (PDF)

S5 Fig. The contribution of genetic and environmental influences on DNA methylation at autosomal sites differs as a function of the variability in DNA methylation level. Shown are estimates of additive genetic effects (A), shared environmental effects (C) and non-shared (or unique) environmental effects (E) plotted as a function of the variability in DNA methylation measured by **A**) the standard deviation (SD) and **B**) the range of the middle 80% of the distribution of DNA methylation levels. In panel **B**, the dashed vertical line indicates the cut-off (5%) used to define DNA methylation sites as being "variable" in this study. (PDF)

S6 Fig. The contribution of genetic and environmental influences on DNA methylation at autosomal sites are not evenly distributed across genic regions. Shown is a line graph depicting the extent to which variation in DNA methylation is influenced by genetic and environmental factors across a canonical gene region. Genetic influences on DNA methylation are highest immediately upstream of the transcription start-site (TSS), and in the region spanning 5 kilobases downstream of the gene coding sequence (red line, panel **A**). Sites located around the TSS are enriched for shared environmental effects (blue line, panel **A**) and show reduced non-shared environmental effects (green line, panel **B**). (PDF)

S7 Fig. The contribution of genetic and environmental influences on DNA methylation at autosomal sites annotated to specific genic features. Shown is a density plot of estimates of **A**) additive genetic, **B**) shared environmental, and **C**) non-shared environmental influences on DNA methylation at autosomal sites stratified by gene feature annotation. (PDF)

S8 Fig. The contribution of genetic and environmental influences on DNA methylation at autosomal sites annotated to specific CpG island features. Shown is a density plot of estimates of **A**) additive genetic, **B**) shared environmental, and **C**) non-shared environmental influences on DNA methylation at sites stratified by CpG island feature annotation. (PDF)

S9 Fig. The influence of genetic and environmental factors on DNAm varies across regulatory features and chromatin states. Violin plots showing the proportion of variance explained by additive genetic factors (A; red), common environmental factors (C; green), and unique environmental factors (E; blue) where DNA methylation sites are stratified by their location in regulatory annotation states as defined by ChromHMM [59] using ENCODE experimental data from the GM12878 cell line. (PDF)

S10 Fig. Sites with intermediate levels of DNAm are associated with larger DNA methylation trait quantitative trait loci (mQTL) effects. Line graph of the moving mean mQTL effect on DNA methylation (measured as the % DNA methylation change per allele; y-axis) as a function of mean DNA methylation (%; x-axis). The gray area indicates the 95% interquantile range for the moving average. (PDF)

(PDF)

S11 Fig. Example of a site at which DNA methylation is highly heritable (A = 96.9%) and associated with genotype at a DNA methylation trait quantitative trait loci (mQTL). Panel A) shows a boxplot of the association between DNA methylation at cg02573566 and genotype at rs11548104 ($P = 5.95 \times 10^{-179}$). Panel B) shows the correlation in DNA methylation at cg02573566 between MZ twins (r = 0.916) and panel C) shows the correlation in DNA methylation at cg02573566 between DZ twins (r = 0.487). (PDF)

S12 Fig. Sites at which DNA methylation is more strongly influenced by genetic factors are more likely to be associated with genotype at a mQTL. Shown is a line graph of the percentage of DNA methylation sites significantly associated with an mQTL variant in our whole blood dataset[29] (y-axis) as a function of increasing cut-offs for estimates of additive genetic (black line), shared environmental (red line) and non-shared environmental (blue line) effects on DNA methylation (x-axis).

(PDF)

S13 Fig. There is a strong correlation between the extent to which inter-individual variation in DNA methylation co-varies across tissues with the influence of additive genetic variation on DNA methylation. Scatterplot of the amount of variance in DNA methylation explained by additive genetic effects (y-axis) against the level of blood-brain covariation in DNA methylation (x-axis) using data from Hannon et al[30] for all sites on the Illumina 450K array. Shown is data for covariation between whole blood and **A**) prefrontal cortex, **B**) entorhinal cortex **C**) superior temporal gyrus and **D**) cerebellum. Color indicates the density of points ranging from yellow (high) to gray (low). PFC = prefrontal cortex, EC = entorhinal cortex, STG = superior temporal gyrus, CER = cerebellum. (PDF)

S14 Fig. Distribution of DNA methylation levels across sites on the X chromosome. A) Shown is a density plot of DNA methylation across sites on the X chromosome stratified by sex. B) Shown is a scatterplot comparing mean DNA methylation at sites across the X-chromosome in females (x-axis) and males (y-axis). (PDF)

S15 Fig. Sex differences for the proportion of variance in DNA methylation explained by additive genetic and environmental influences for sites on the X chromosome. Shown are density plots of estimates of additive genetic effects (A), shared environmental effects (C) and non-shared (or unique) environmental effects (E) stratified by sex (red = females, green = males). (PDF)

S16 Fig. The contribution of genetic and environmental influences on DNA methylation at sites on the X-chromosome differs as a function of mean DNA methylation with notable differences between males and females. Shown for A) males and B) females are estimates of additive genetic effects (A), shared environmental effects (C) and non-shared (or unique) environmental effects (E) plotted as a function of average DNA methylation level. (PDF)

S17 Fig. The contribution of genetic and environmental influences on DNA methylation at sites on the X-chromosome differs as a function of the variability in DNA methylation level, with notable differences between males and females. Shown are estimates of additive genetic effects (A), shared environmental effects (C) and unshared (or unique) environmental effects (E) against probe variability. Panels A and B show how genetic and environmental influences differ as a function of the standard deviation (SD) in DNA methylation in males and females, respectively. Panels C and D show how genetic and environmental influences differ as a function of the distribution of DNA methylation levels in males and females, respectively. The dashed vertical line indicates the cut-off of 5% used to define probes as being "variable". (PDF)

S18 Fig. The contribution of genetic and environmental influences on DNA methylation at sites on the X-chromosome is modestly correlated between males and females. Shown are scatterplots of the **A**) additive genetic, **B**) shared environmental, and **C**) non-shared environmental contribution to DNA methylation for sites on the X chromosome in female (x-axis) and male (y-axis) twin pairs.

(PDF)

S19 Fig. An example of a site (cg00195237) on the X chromosome at which DNA methylation is strongly influenced by additive genetic factors in females (A = 56.8%) but not males (A = 6.70%). The scatterplots show DNA methylation values in A) female MZ, B) female DZ, D) male MZ, and E) male DZ twin pairs. Each point represents an individual twin-pair. At this site, the twin correlation of DNA methylation is notably higher in female MZ twins (r = 0.569) compared to female DZ twins (r = 0.15), whereas the correlations for male MZ twins (r = 0.0686) and male DZ twins (r = 0.0289) are similar. Results from structural equation modelling are presented as stacked bar-plots for C) female and F) male twin-pairs respectively, highlighting higher genetic effects on DNA methylation at this site in females than males. (PDF)

S20 Fig. An example of a site (cg19782749) on the X chromosome at which DNA methylation is strongly influenced by additive genetic factors in males (A = 58.9%) but not females (A = 3.76%). The scatterplots show DNA methylation values in A) female MZ, B) female DZ, D) male MZ, and E) male DZ twin pairs. Each point represents an individual twin-pair. At this site, the correlation of DNA methylation is notably higher in male MZ twins (r = 0.777) compared to female DZ twins (r = 0.536), whereas the correlations for male MZ twins (r = 0.405) and male DZ twins (r = 0.378) are similar. Results from structural equation modelling are presented as stacked bar-plots for C) female and F) male twin-pairs respectively, highlighting higher heritability in females than males. (TIF)

S21 Fig. The proportion of variance in DNA methylation explained by additive genetic and environmental influences for sites on the X chromosome. Shown are density plots of estimates of additive genetic effects (A), shared environmental effects (C) and non-shared (or

unique) environmental effects (E) stratified by sex and within females stratified by sites located in the transcription start site or 5'UTR of genes that escape XCI (red = females, blue = females sites that escape X chromosome inactivation, green = males). (PDF)

S22 Fig. Twin-pair correlations for estimates of DNA methylation age and blood cell composition derived from DNA methylation data. Shown are co-twin correlations for **A**) DNA methylation age, **B**) estimated plasma blast abundance, **C**) estimated CD8+CD28-CD45RA-T cell abundance, **D**) estimated naïve CD8 T cell abundance, **E**) estimated naïve CD4 T cell abundance (all derived using the online Epigenetic Clock software[<u>32</u>]), **F**) estimated CD8 T cell proportion, **G**) estimated CD4 T cell proportion, **H**) estimated natural killer cell proportion, **I**) estimated B cell proportion, **J**) estimated monocyte proportion, and **K**) estimated granulocyte proportion (all derived using the Houseman algorithm[<u>33</u>, <u>34</u>]). Panels on the left show correlations for monozygotic (MZ) twin pairs and panels on the right show correlations for dizygotic (DZ) twin pairs. (PDF)

S23 Fig. The contribution of additive genetic and environmental influences to age and blood cell-count estimates derived from the DNA methylation data. AAR = age acceleration residual derived from the DNA methylation age clock. (PDF)

S24 Fig. Effect sizes at DNA methylation sites associated with tobacco smoking in the Erisk cohort overlap with those previously identified in adult cohorts. A) The mean difference between current smokers and never smokers from the E-risk cohort (x-axis) against a similar study in adults taken from Joehanes et al[44] (y-axis). B) Shown is the correlation of the signed log10 P-values from a comparison between current smokers and never smokers from the E-risk cohort (x-axis) against a similar study in adults taken from Joehanes et al[44] (y-axis).

(PDF)

S25 Fig. Examples of DNA methylation sites associated with smoking that are influenced by both additive genetic and environmental factors. Scatterplot of DNA methylation values at cg05575921 for A) monozygotic (MZ) twin pairs and B) dizygotic (DZ) twin pairs, and cg26703534 for C) MZ twin pairs and D) DZ twin pairs. Colors depict the concordance for current smoking status in each twin-pair. (TIF)

S26 Fig. DNA methylation at sites associated with body mass index (BMI) is influenced by additive genetic factors. Density plots for estimates of **A**) additive genetic effects (A), **B**) shared environmental effects (C), and **C**) non-shared environmental effects (E) at 176 differentially methylated positions (DMPs) recently associated with BMI (green)[<u>47</u>]. (PDF)

Author Contributions

Conceptualization: Eilis Hannon, Avshalom Caspi, Jonathan Mill.

Data curation: Eilis Hannon, Joe Burrage.

Formal analysis: Eilis Hannon, Olivia Knox, Karen Sugden.

Funding acquisition: Louise Arseneault, Terrie E. Moffitt, Avshalom Caspi, Jonathan Mill.

Investigation: Eilis Hannon, Avshalom Caspi, Jonathan Mill.

Methodology: Eilis Hannon, Daniel W. Belsky, David L. Corcoran.

Project administration: Louise Arseneault, Terrie E. Moffitt, Avshalom Caspi.

Supervision: Avshalom Caspi, Jonathan Mill.

- Writing original draft: Eilis Hannon, Avshalom Caspi, Jonathan Mill.
- Writing review & editing: Eilis Hannon, Olivia Knox, Karen Sugden, Joe Burrage, Chloe C. Y. Wong, Daniel W. Belsky, David L. Corcoran, Louise Arseneault, Terrie E. Moffitt, Avshalom Caspi, Jonathan Mill.

References

- van Dongen J, Slagboom PE, Draisma HH, Martin NG, Boomsma DI. The continuing value of twin studies in the omics era. Nat Rev Genet. 2012; 13(9):640–53. <u>https://doi.org/10.1038/nrg3243</u> PMID: <u>22847273</u>.
- Polderman TJ, Benyamin B, de Leeuw CA, Sullivan PF, van Bochoven A, Visscher PM, et al. Metaanalysis of the heritability of human traits based on fifty years of twin studies. Nat Genet. 2015; 47 (7):702–9. Epub 2015/05/18. <u>https://doi.org/10.1038/ng.3285</u> PMID: <u>25985137</u>.
- Castillo-Fernandez JE, Spector TD, Bell JT. Epigenetics of discordant monozygotic twins: implications for disease. Genome Med. 2014; 6(7):60. Epub 2014/07/31. <u>https://doi.org/10.1186/s13073-014-0060-</u> z PMID: 25484923; PubMed Central PMCID: PMCPMC4254430.
- Mill J, Heijmans BT. From promises to practical strategies in epigenetic epidemiology. Nat Rev Genet. 2013; 14(8):585–94. <u>https://doi.org/10.1038/nrg3405</u> PMID: <u>23817309</u>.
- Birney E, Smith GD, Greally JM. Epigenome-wide Association Studies and the Interpretation of Disease -Omics. Plos Genet. 2016; 12(6):e1006105. Epub 2016/06/23. <u>https://doi.org/10.1371/journal.pgen.</u> 1006105 PMID: 27336614; PubMed Central PMCID: PMCPMC4919098.
- Elliott HR, Tillin T, McArdle WL, Ho K, Duggirala A, Frayling TM, et al. Differences in smoking associated DNA methylation patterns in South Asians and Europeans. Clin Epigenetics. 2014; 6(1):4. https://doi.org/10.1186/1868-7083-6-4 PMID: 24485148; PubMed Central PMCID: PMCPMC3915234.
- Tsaprouni LG, Yang TP, Bell J, Dick KJ, Kanoni S, Nisbet J, et al. Cigarette smoking reduces DNA methylation levels at multiple genomic loci but the effect is partially reversible upon cessation. Epigenetics. 2014; 9(10):1382–96. https://doi.org/10.4161/15592294.2014.969637 PMID: 25424692.
- Joubert BR, Felix JF, Yousefi P, Bakulski KM, Just AC, Breton C, et al. DNA Methylation in Newborns and Maternal Smoking in Pregnancy: Genome-wide Consortium Meta-analysis. Am J Hum Genet. 2016; 98(4):680–96. <u>https://doi.org/10.1016/j.ajhg.2016.02.019</u> PMID: <u>27040690</u>; PubMed Central PMCID: PMCPMC4833289.
- Pauwels S, Ghosh M, Duca RC, Bekaert B, Freson K, Huybrechts I, et al. Maternal intake of methylgroup donors affects DNA methylation of metabolic genes in infants. Clin Epigenetics. 2017; 9:16. Epub 2017/02/07. <u>https://doi.org/10.1186/s13148-017-0321-y</u> PMID: <u>28191262</u>; PubMed Central PMCID: PMCPMC5297118.
- Tobi EW, Goeman JJ, Monajemi R, Gu H, Putter H, Zhang Y, et al. DNA methylation signatures link prenatal famine exposure to growth and metabolism. Nat Commun. 2014; 5:5592. Epub 2014/11/26. https://doi.org/10.1038/ncomms6592 PMID: 25424739; PubMed Central PMCID: PMCPMC4246417.
- Kumsta R, Marzi SJ, Viana J, Dempster EL, Crawford B, Rutter M, et al. Severe psychosocial deprivation in early childhood is associated with increased DNA methylation across a region spanning the transcription start site of CYP2E1. Transl Psychiatry. 2016; 6(6):e830. Epub 2016/06/07. <u>https://doi.org/10. 1038/tp.2016.95</u> PMID: <u>27271856</u>; PubMed Central PMCID: PMCPMC4931613.
- Essex MJ, Boyce WT, Hertzman C, Lam LL, Armstrong JM, Neumann SM, et al. Epigenetic vestiges of early developmental adversity: childhood stress exposure and DNA methylation in adolescence. Child Dev. 2013; 84(1):58–75. Epub 2011/09/02. <u>https://doi.org/10.1111/j.1467-8624.2011.01641.x</u> PMID: 21883162; PubMed Central PMCID: PMCPMC3235257.
- Hannon E, Spiers H, Viana J, Pidsley R, Burrage J, Murphy TM, et al. Methylation QTLs in the developing brain and their enrichment in schizophrenia risk loci. Nat Neurosci. 2015. <u>https://doi.org/10.1038/nn.</u> <u>4182</u> PMID: <u>26619357</u>.
- 14. Wagner JR, Busche S, Ge B, Kwan T, Pastinen T, Blanchette M. The relationship between DNA methylation, genetic and expression inter-individual variation in untransformed human fibroblasts. Genome

Biol. 2014; 15(2):R37. <u>https://doi.org/10.1186/gb-2014-15-2-r37</u> PMID: <u>24555846</u>; PubMed Central PMCID: PMCPMC4053980.

- Gibbs JR, van der Brug MP, Hernandez DG, Traynor BJ, Nalls MA, Lai SL, et al. Abundant quantitative trait loci exist for DNA methylation and gene expression in human brain. Plos Genet. 2010; 6(5): e1000952. <u>https://doi.org/10.1371/journal.pgen.1000952</u> PMID: <u>20485568</u>; PubMed Central PMCID: PMCPMC2869317.
- Olsson AH, Volkov P, Bacos K, Dayeh T, Hall E, Nilsson EA, et al. Genome-wide associations between genetic and epigenetic variation influence mRNA expression and insulin secretion in human pancreatic islets. Plos Genet. 2014; 10(11):e1004735. <u>https://doi.org/10.1371/journal.pgen.1004735</u> PMID: 25375650; PubMed Central PMCID: PMCPMC4222689.
- Gutierrez-Arcelus M, Lappalainen T, Montgomery SB, Buil A, Ongen H, Yurovsky A, et al. Passive and active DNA methylation and the interplay with genetic variation in gene regulation. Elife. 2013; 2: e00523. Epub 2013/06/04. <u>https://doi.org/10.7554/eLife.00523</u> PMID: <u>23755361</u>; PubMed Central PMCID: PMCPMC3673336.
- Grundberg E, Meduri E, Sandling JK, Hedman AK, Keildson S, Buil A, et al. Global analysis of DNA methylation variation in adipose tissue from twins reveals links to disease-associated variants in distal regulatory elements. Am J Hum Genet. 2013; 93(5):876–90. <u>https://doi.org/10.1016/j.ajhg.2013.10.004</u> PMID: <u>24183450</u>; PubMed Central PMCID: PMCPMC3824131.
- Bell JT, Tsai PC, Yang TP, Pidsley R, Nisbet J, Glass D, et al. Epigenome-wide scans identify differentially methylated regions for age and age-related phenotypes in a healthy ageing population. Plos Genet. 2012; 8(4):e1002629. <u>https://doi.org/10.1371/journal.pgen.1002629</u> PMID: <u>22532803</u>; PubMed Central PMCID: PMCPMC3330116.
- Gordon L, Joo JE, Powell JE, Ollikainen M, Novakovic B, Li X, et al. Neonatal DNA methylation profile in human twins is specified by a complex interplay between intrauterine environmental and genetic factors, subject to tissue-specific influence. Genome Res. 2012; 22(8):1395–406. <u>https://doi.org/10.1101/</u> gr.136598.111 PMID: 22800725; PubMed Central PMCID: PMC3409253.
- van Dongen J, Nivard MG, Willemsen G, Hottenga JJ, Helmer Q, Dolan CV, et al. Genetic and environmental influences interact with age and sex in shaping the human methylome. Nat Commun. 2016; 7:11115. <u>https://doi.org/10.1038/ncomms11115</u> PMID: <u>27051996</u>; PubMed Central PMCID: PMCPMC4820961.
- Moffitt TE, Team E-RS. Teen-aged mothers in contemporary Britain. J Child Psychol Psychiatry. 2002; 43(6):727–42. PMID: <u>12236608</u>.
- McRae AF, Powell JE, Henders AK, Bowdler L, Hemani G, Shah S, et al. Contribution of genetic variation to transgenerational inheritance of DNA methylation. Genome Biol. 2014; 15(5):R73. <u>https://doi.org/10.1186/gb-2014-15-5-r73</u> PMID: <u>24887635</u>; PubMed Central PMCID: PMCPMC4072933.
- Jones PA. Functions of DNA methylation: islands, start sites, gene bodies and beyond. Nat Rev Genet. 2012; 13(7):484–92. Epub 2012/05/29. <u>https://doi.org/10.1038/nrg3230</u> PMID: <u>22641018</u>.
- Ball MP, Li JB, Gao Y, Lee JH, LeProust EM, Park IH, et al. Targeted and genome-scale strategies reveal gene-body methylation signatures in human cells. Nat Biotechnol. 2009; 27(4):361–8. Epub 2009/03/29. <u>https://doi.org/10.1038/nbt.1533</u> PMID: <u>19329998</u>; PubMed Central PMCID: PMCPMC3566772.
- Maunakea AK, Nagarajan RP, Bilenky M, Ballinger TJ, D'Souza C, Fouse SD, et al. Conserved role of intragenic DNA methylation in regulating alternative promoters. Nature. 2010; 466(7303):253–7. <u>https:// doi.org/10.1038/nature09165</u> PMID: 20613842; PubMed Central PMCID: PMCPMC3998662.
- Drong AW, Nicholson G, Hedman AK, Meduri E, Grundberg E, Small KS, et al. The presence of methylation quantitative trait loci indicates a direct genetic influence on the level of DNA methylation in adipose tissue. PLoS One. 2013; 8(2):e55923. <u>https://doi.org/10.1371/journal.pone.0055923</u> PMID: <u>23431366</u>; PubMed Central PMCID: PMCPMC3576415.
- Gibbs JR, van der Brug MP, Hernandez DG, Traynor BJ, Nalls MA, Lai SL, et al. Abundant Quantitative Trait Loci Exist for DNA Methylation and Gene Expression in Human Brain. Plos Genet. 2010; 6(5). doi: ARTN e1000952 <u>https://doi.org/10.1371/journal.pgen.1000952</u> PubMed PMID: ISI:000278557300022. PMID: 20485568
- Hannon E, Dempster E, Viana J, Burrage J, Smith AR, Macdonald R, et al. An integrated genetic-epigenetic analysis of schizophrenia: evidence for co-localization of genetic associations and differential DNA methylation. Genome biology. 2016; 17(1):176. <u>https://doi.org/10.1186/s13059-016-1041-x</u> PMID: <u>27572077</u>; PubMed Central PMCID: PMCPMC5004279.
- Hannon E, Lunnon K, Schalkwyk L, Mill J. Interindividual methylomic variation across blood, cortex, and cerebellum: implications for epigenetic studies of neurological and neuropsychiatric phenotypes. Epigenetics. 2015; 10(11):1024–32. <u>https://doi.org/10.1080/15592294.2015.1100786</u> PMID: <u>26457534</u>.

- Tukiainen T, Villani AC, Yen A, Rivas MA, Marshall JL, Satija R, et al. Landscape of X chromosome inactivation across human tissues. Nature. 2017; 550(7675):244–8. <u>https://doi.org/10.1038/</u> nature24265 PMID: 29022598; PubMed Central PMCID: PMCPMC5685192.
- 32. Horvath S. DNA methylation age of human tissues and cell types. Genome Biol. 2013; 14(10):R115. <u>https://doi.org/10.1186/gb-2013-14-10-r115</u> PMID: <u>24138928</u>; PubMed Central PMCID: PMCPMC4015143.
- 33. Houseman EA, Accomando WP, Koestler DC, Christensen BC, Marsit CJ, Nelson HH, et al. DNA methylation arrays as surrogate measures of cell mixture distribution. BMC Bioinformatics. 2012; 13:86. <u>https://doi.org/10.1186/1471-2105-13-86</u> PMID: <u>22568884</u>; PubMed Central PMCID: PMCPMC3532182.
- Koestler DC, Christensen B, Karagas MR, Marsit CJ, Langevin SM, Kelsey KT, et al. Blood-based profiles of DNA methylation predict the underlying distribution of cell types: a validation analysis. Epigenetics. 2013; 8(8):816–26. <u>https://doi.org/10.4161/epi.25430</u> PMID: <u>23903776</u>; PubMed Central PMCID: PMCPMC3883785.
- Chen BH, Marioni RE, Colicino E, Peters MJ, Ward-Caviness CK, Tsai PC, et al. DNA methylationbased measures of biological age: meta-analysis predicting time to death. Aging (Albany NY). 2016; 8 (9):1844–65. <u>https://doi.org/10.18632/aging.101020</u> PMID: <u>27690265</u>; PubMed Central PMCID: PMCPMC5076441.
- 36. Horvath S, Gurven M, Levine ME, Trumble BC, Kaplan H, Allayee H, et al. An epigenetic clock analysis of race/ethnicity, sex, and coronary heart disease. Genome Biol. 2016; 17(1):171. Epub 2016/08/11. https://doi.org/10.1186/s13059-016-1030-0 PMID: 27511193; PubMed Central PMCID: PMCPMC4980791.
- Horvath S, Langfelder P, Kwak S, Aaronson J, Rosinski J, Vogt TF, et al. Huntington's disease accelerates epigenetic aging of human brain and disrupts DNA methylation levels. Aging (Albany NY). 2016; 8 (7):1485–512. <u>https://doi.org/10.18632/aging.101005</u> PMID: <u>27479945</u>; PubMed Central PMCID: PMCPMC4993344.
- Aguirre-Gamboa R, Joosten I, Urbano PC, van der Molen RG, van Rijssen E, van Cranenbroek B, et al. Differential Effects of Environmental and Genetic Factors on T and B Cell Immune Traits. Cell Rep. 2016; 17(9):2474–87. Epub 2016/11/03. <u>https://doi.org/10.1016/j.celrep.2016.10.053</u> PMID: <u>27818087</u>; PubMed Central PMCID: PMCPMC5130901.
- Orrù V, Steri M, Sole G, Sidore C, Virdis F, Dei M, et al. Genetic variants regulating immune cell levels in health and disease. Cell. 2013; 155(1):242–56. <u>https://doi.org/10.1016/j.cell.2013.08.041</u> PMID: <u>24074872</u>.
- **40.** Brodin P, Jojic V, Gao T, Bhattacharya S, Angel CJ, Furman D, et al. Variation in the human immune system is largely driven by non-heritable influences. Cell. 2015; 160(1–2):37–47. https://doi.org/10.1016/j.cell.2014.12.020 PMID: https://doi.0216/j.cell.2014.12.020 PMID: https://doi.org/10.1016/j.cell.2014.12.020 PMID: https://doi.org/10.1016/j.cell.2014.12.020 PMID: >>>>>>>>>>>>>>>>>>>>>>>>>>>>>>
- Mangino M, Roederer M, Beddall MH, Nestle FO, Spector TD. Innate and adaptive immune traits are differentially affected by genetic and environmental factors. Nat Commun. 2017; 8:13850. Epub 2017/ 01/05. <u>https://doi.org/10.1038/ncomms13850</u> PMID: <u>28054551</u>; PubMed Central PMCID: PMCPMC5227062.
- 42. Shenker NS, Polidoro S, van Veldhoven K, Sacerdote C, Ricceri F, Birrell MA, et al. Epigenome-wide association study in the European Prospective Investigation into Cancer and Nutrition (EPIC-Turin) identifies novel genetic loci associated with smoking. Hum Mol Genet. 2013; 22(5):843–51. <u>https://doi.org/10.1093/hmg/dds488 PMID: 23175441</u>.
- Sun YV, Smith AK, Conneely KN, Chang Q, Li W, Lazarus A, et al. Epigenomic association analysis identifies smoking-related DNA methylation sites in African Americans. Hum Genet. 2013; 132 (9):1027–37. <u>https://doi.org/10.1007/s00439-013-1311-6</u> PMID: <u>23657504</u>; PubMed Central PMCID: PMCPMC3744600.
- Joehanes R, Just AC, Marioni RE, Pilling LC, Reynolds LM, Mandaviya PR, et al. Epigenetic Signatures of Cigarette Smoking. Circ Cardiovasc Genet. 2016; 9(5):436–47. Epub 2016/09/20. <u>https://doi.org/10. 1161/CIRCGENETICS.116.001506</u> PMID: 27651444; PubMed Central PMCID: PMCPMC5267325.
- Vink JM, Willemsen G, Boomsma DI. Heritability of smoking initiation and nicotine dependence. Behav Genet. 2005; 35(4):397–406. https://doi.org/10.1007/s10519-004-1327-8 PMID: 15971021.
- 46. Vink JM, Smit AB, de Geus EJ, Sullivan P, Willemsen G, Hottenga JJ, et al. Genome-wide association study of smoking initiation and current smoking. Am J Hum Genet. 2009; 84(3):367–79. <u>https://doi.org/ 10.1016/j.ajhg.2009.02.001</u> PMID: <u>19268276</u>; PubMed Central PMCID: PMCPMC2667987.
- Wahl S, Drong A, Lehne B, Loh M, Scott WR, Kunze S, et al. Epigenome-wide association study of body mass index, and the adverse outcomes of adiposity. Nature. 2017; 541(7635):81–6. Epub 2016/ 12/21. <u>https://doi.org/10.1038/nature20784</u> PMID: <u>28002404</u>; PubMed Central PMCID: PMCPMC5570525.

- Pidsley R, Zotenko E, Peters TJ, Lawrence MG, Risbridger GP, Molloy P, et al. Critical evaluation of the Illumina MethylationEPIC BeadChip microarray for whole-genome DNA methylation profiling. Genome Biol. 2016; 17(1):208. Epub 2016/10/07. <u>https://doi.org/10.1186/s13059-016-1066-1</u> PMID: <u>27717381</u>; PubMed Central PMCID: PMCPMC5055731.
- Odgers CL, Caspi A, Bates CJ, Sampson RJ, Moffitt TE. Systematic social observation of children's neighborhoods using Google Street View: a reliable and cost-effective method. J Child Psychol Psychiatry. 2012; 53(10):1009–17. <u>https://doi.org/10.1111/j.1469-7610.2012.02565.x</u> PMID: <u>22676812</u>; PubMed Central PMCID: PMCPMC3537178.
- Marzi SJ, Sugden K, Arseneault L, Belsky DW, Burrage J, Corcoran DL, et al. Analysis of DNA Methylation in Young People: Limited Evidence for an Association Between Victimization Stress and Epigenetic Variation in Blood. Am J Psychiatry. 2018:appiajp201717060693. Epub 2018/01/12. <u>https://doi.org/10.1176/appi.ajp.2017.17060693</u> PMID: 29325449.
- 51. Davis S, Du P, Bilke S, Triche J, Bootwalla M. methylumi: Handle Illumina methylation data. R package version 2.14.0. 2015.
- 52. Pidsley R, Wong CCY, Volta M, Lunnon K, Mill J, Schalkwyk LC. A data-driven approach to preprocessing Illumina 450K methylation array data. Bmc Genomics. 2013; 14. doi: Unsp 293. <u>https://doi.org/10.</u> <u>1186/1471-2164-14-293</u> PubMed PMID: ISI:000319819700001. PMID: <u>23631413</u>
- Chen YA, Lemire M, Choufani S, Butcher DT, Grafodatskaya D, Zanke BW, et al. Discovery of crossreactive probes and polymorphic CpGs in the Illumina Infinium HumanMethylation450 microarray. Epigenetics. 2013; 8(2):203–9. <u>https://doi.org/10.4161/epi.23470</u> PMID: <u>23314698</u>; PubMed Central PMCID: PMCPMC3592906.
- Price ME, Cotton AM, Lam LL, Farré P, Emberly E, Brown CJ, et al. Additional annotation enhances potential for biologically-relevant analysis of the Illumina Infinium HumanMethylation450 BeadChip array. Epigenetics Chromatin. 2013; 6(1):4. <u>https://doi.org/10.1186/1756-8935-6-4</u> PMID: <u>23452981</u>; PubMed Central PMCID: PMCPMC3740789.
- 55. Boker S, Neale M, Maes H, Wilde M, Spiegel M, Brick T, et al. OpenMx: An Open Source Extended Structural Equation Modeling Framework. Psychometrika. 2011; 76(2):306–17. <u>https://doi.org/10.1007/s11336-010-9200-6</u> PMID: <u>23258944</u>; PubMed Central PMCID: PMCPMC3525063.
- 56. Neale MC, Hunter MD, Pritikin JN, Zahery M, Brick TR, Kirkpatrick RM, et al. OpenMx 2.0: Extended Structural Equation and Statistical Modeling. Psychometrika. 2016; 81(2):535–49. <u>https://doi.org/10. 1007/s11336-014-9435-8</u> PMID: 25622929; PubMed Central PMCID: PMCPMC4516707.
- 57. Croissant Y, Millo G. Panel Data Econometrics in R: The plm Package. Journal of Statistical Software. 2008; 27(2).
- Zeileis A. Econometric Computing with HC and HAC Covariance Matrix Estimators. Journal of Statistical Software. 2004; 11(10):1–17.
- Ernst J, Kellis M. ChromHMM: automating chromatin-state discovery and characterization. Nat Methods. 2012; 9(3):215–6. Epub 2012/02/28. <u>https://doi.org/10.1038/nmeth.1906</u> PMID: <u>22373907</u>; PubMed Central PMCID: PMCPMC3577932.